vc/27-Nov-03
Week 48 starting 24 nov 03

SM18 Operation SCHEDULE Meeting (Tuesdays)

Schedule for this week : 
	BENCH
	CURRENT

MAGNET
	STATUS 

Today(25/11/03)
	Forward look,   & Next Magnet 

	TBA1
	NOT Active
	
	Bench de-activated – preparation for D bench

“Park” here 2001 ( moved from D bench )

	TBA2
	1028
	Pump down?
	

	TBC1
	1029
	cold
	3021 next ?

	TBC2
	SSS04
	
	

	TBE1
	2027
	 cold
	

	TBE2
	2029
	disconnection
	1032 next ?

	TBF1
	3065
	Connection
	Arrived yesterday

	TBF2
	1027
	cooldown
	


 In pipeline Next 3015

Test Coordinator this week:  Marco B;  MTM  Tech. on call:  (16-3602) : 

· Situation From Cryogenics:  J-P Lamboy or ACR Rep.

· Other General Issues
1. Hardware Recog. System implementation/use

2. AOB

Problems of Last week

	Date
	Problem
	Type
	Application/soft.
	Action/advise

	Tuesday 18_A


	TBE2: IAP@warm (all HF)

Trigger generate two files. qa01 was faulty


	MTM
	IAP
	

	Tuesday 18_A


	TBE2: QH measurements: MCDO is showing 1.16 ohms


	MTM
	
	Non conformity issued

	Tuesday 18_N


	TBF2 : Problem with the TRU 

Change of CPU no results
	MTM/AB-CO
	
	Solved Wednesday by changing a crate.

 

	Tuesday 18_N


	TBF2: Full load line aborted at 7800 A due to cryo OK lost.

Problem during recuperation

7hrs lost
	ACR
	
	Repeated.

Bad parameters for cryo control 

	Tuesday 18_N


	TBA2 : RRR could not be performed because of no stable value of Temperature.
	ACR
	
	 

	Tuesday 18_N


	TBA2 : Sun MTA 1 hang: impossible to launch the warm up.
	
	
	 

	Wednesday 19_M


	TBA2 : LHC cycle failed 

Cryo OK lost.
	ACR
	
	Repeated.

 

	Wednesday 19_A


	TBE1 : Problem on Tanticryo due to unconnected cable.
	ICS
	
	 

	Wednesday 19_A


	TBE2 : 78 faulty triggers files deleted from dacqdisk
	MTM
	
	 

	Thursday 20_M


	TBA2 (3022): QH measurements at warm forgotten.


	MTM
	
	Instructions given by Nick-

QH resistance measured outside the SM18

	Thursday 20_A


	TBA2 (3022) : Problem with provoked quench (all LF).

Before the change of shift, change of protection LF to HF. Next shift believed that quench was wrong because the switch was not in correct position 


	MTM
	
	See comments by Prashant.

 

	Thursday 20_N

Till Monday 23 _M


	TBF1 (3059): HV test at cold failed (1850 V dipole vs ground)


	MTM
	
	Identification of the circuit and investigation step by step.

Change of the IFS box…

 Test with portable megger…

Check By A.Siemko

Warm up the magnet.

Non conformity created by test coordinator.

	Friday 21_M


	TBF1: HV test :problem to read the data.
	AB/CO
	HV
	Application transferred to sun MTA 26 .



	Friday 21_


	TBE2: Water problem on power supply

“vane” for cable cooling closed .
	SL/PO
	
	 

	Saturday 22_N


	TBC1 (3059): IAP@cold without shaft

Error in V_T cat et VT_ano


	MTM
	
	I
 

	Sunday 23_M


	TBC1: Slow power abort could not be launched.

Power supply could not start.

Interlock due to Tprobe in the current leads out of use.
	ACR
	
	Magnet warmed and then cooled down again.

 

	Sunday 23_M


	TBE1: Tcoil testing

D1/S01 not OK
	MTM
	
	 

	Sunday 23_A

Sunday 23_N


	TBE1: MM0: Problem in the 24 V power supply of motor 1 (card in the control module loose)

Problem in b11 aperture 1


	MTM
	MMP
	For this aperture measurement with 13 sectors: failed

Other strategy chosen: measurements from sector 2 to sector 13 . Problem with the implementation of the shift.

Solved by Guy/Ewald.

	Monday  24_A


	TBF1: IAP @warm PLC errors

Problem with rotary switch in CC position
	MTM
	IAP
	 Giorgio called

	Monday  24_A


	TBA2: (1028) VT resistance not OK
	MTM
	
	 Non conformity opened.


